Prepared for:



# Waste Connections of Osceola County, LLC.

1501 Omni Way St. Cloud, Florida 34773

# OPERATION PLAN J.E.D. SOLID WASTE MANAGEMENT FACILITY Rev. 03

Prepared by:

# Geosyntec<sup>D</sup>

consultants

12802 Tampa Oaks Blvd., Suite 151 Tampa, Florida 33637 Authorization Number: 4321

Project Number FL3616

June 2019

Revised by Golder Associates Inc. June 2021 Revised by Golder Associates IISA Inc. October 2022

No. 81578

STATE OF

STATE

# TABLE OF CONTENTS

1	INT	RODU	CTION	1		
	1.1	Terms	of Reference	1		
	1.2	_	se and Scope of the Operation Plan			
	1.3	Operat	Operation Plan Organization			
2	LA	NDFIL	L PERSONNEL AND FACILITIES	2		
	2.1	Personnel Requirements		2		
		2.1.1	Numbers and Types.			
		2.1.2	Employee Training	2		
	2.2	Landfi	ll Entrance Procedures	3		
		2.2.1	Hours and Days of Operation	3		
		2.2.2	Processing Customers			
		2.2.3	Public Use			
		2.2.4	Vehicle Inspection			
	2.3	Traffic	Routing			
		2.3.1	Access Points/Signs			
		2.3.2	On-Site Traffic Flow			
	2.4					
	2.5	Communication Facilities		6		
3	LANDFILL OPERATIONS					
	3.1	Basic Landfilling Procedures				
		3.1.1	Method of Operations	7		
		3.1.2	Working Face Practices.	8		
			3.1.2.1 Start-Up and First Lift	8		
			3.1.2.2 Subsequent Lifts	9		
		3.1.3	Filling Procedures	10		
			3.1.3.1 Width of Working Face	10		
			3.1.3.2 Unloading	10		
			3.1.3.3 Pushing, Spreading, and Compacting			
			3.1.3.4 Daily Clean-Up	1		
		3.1.4	Cover			
			3.1.4.1 Stockpiling			
			3.1.4.2 Application and Phasing of Cover Materials			
	3.2		ment			
		3.2.1	Primary Equipment	12		

			4.6.1.2 Control of Litter with Litter Fences	22
		4.6.2	Buffer Maintenance	23
		4.6.3	Dust Control	
		4.6.4	Vector Control	23
		4.6.5	Noise Control	24
		4.6.6	Recordkeeping	24
5	EM	ERGE	NCY CONTINGENCY PLAN	26
	5.1	Introd	uction	26
	5.2	Fire Control Plan		
		5.2.1	When Fire Occurs	27
		5.2.2	"Hot Load" Procedures	27
		5.2.3	Fire Extinguishers and First Aid Kits.	28
	5.3	Accid	ent or Injury	28
		5.3.1	When an Injury Occurs	28
		5.3.2	Procedures After an Accident	28
	5.4	Releas	se of Contamination to Environment (Remedial Response)	29
		5.4.1	Response	29
		5.4.2	Follow-Up	30
	5.5	Hazardous or Other Unauthorized Materials		30
		5.5.1	The Observer	30
			5.5.1.1 Known or Suspected Hazardous Material	30
			5.5.1.2 Other Unauthorized Material	31
		5.5.2	Landfill Site Management	31
		5.5.3	Non-Discharged Load	31
	5.6	Uncoo	operative Customers	32
	5.7	Inclen	nent Weather	32
		5.7.1	Operation in Wet Weather	32
		5.7.2	Preparation for Inclement Weather	32
		5.7.3	Preparation for Severe Weather or Hurricanes	33
	5.8	Problems Affecting the Leachate Collection and Removal Systems		34
		5.8.1	Interruption of Power Service to the Landfill	34
		5.8.2	Interruption of Flow to Leachate Storage Facility	35
		5.8.3	Primary Leachate Sump Alarm Level Switch	
		5.8.4	Managing Hazardous Leachate	35
6	SAI	ETY I	PLAN	36
	6.1	Emers	gency Procedures	36

				Geosyntec Consultants			
			ety Practices				
	6.3	• •	pmentules				
_							
7	_	OSURE PLA		39			
	7.1		Sustan Decima				
	7.2 7.3		System Designter Drainage System				
	7.4		System Components				
			etation				
		U	etative and Cap Protective Layer				
		7.4.3 Geo	composite Drainage Layer	40			
		7.4.4 Geo	membrane	41			
	7.5	Erosion Min	nimization	41			
8	BORROW AREA OPERATIONS						
	8.1	Overview		42			
			tion				
	8.3	Dry Excava	tion	43			
LIS	T O	F TABLES					
Tab	ole 1	Personnel Re	quirements	44			
Tab	ole 2	Heavy Equip	ment Requirements <sup>(1)</sup> ······	45			
Tab	ole 3	Heavy Equip	ment Rental Companies	46			
Tab	ole 4	Operator Pro	tective Equipment	47			
ΑT	TAC	HMENTS					
AT	TACI	HMENT A:	TRAINING PLAN				
AT	TACI	HMENT B:	WASTE INSPECTION PLAN				
AT	TACI	HMENT C:	WASTE TIRE STORAGE AND PROCESSI	NG PLAN			
AT	TACI	HMENT D:	LFGTE FACILITY OPERATION PLAN				
AT	TACI	HMENT E:	OPERATION PLAN FOR LEACHATE REC	CIRCULATION			
AT	ATTACHMENT F: WASTE SOLIDIFICATION OPERATION PLAN						
AT	TACI	HMENT G:	EMERGENCY RESPONSE AGENCIES				
AT	TACI	HMENT H:	LEACHATE EVAPORATION SYSTEM				

#### 1 INTRODUCTION

# 1.1 Terms of Reference

This Operation Plan has been prepared by Geosyntec Consultants (Geosyntec) on behalf of Waste Connections of Osceola County, LLC (WCOC) for a Class I landfill known as the J.E.D. Solid Waste Management (JED) facility. The Operation Plan for the JED facility has been prepared to comply with the requirements of Chapter 62-701, Florida Administrative Code (F.A.C.).

# 1.2 Purpose and Scope of the Operation Plan

The Operation Plan provides a detailed description of the daily operations at the JED facility, including contingency operations as required by subparagraphs 62- 701.320(7)(e)1,2, and Rule 62-701.500, F.A.C. The primary purpose of the Operation Plan is to describe the framework to operate and manage the JED facility so that the landfill is operated and maintained in a condition that protects the public health and the environment. This Operation Plan also provides a description of borrow area operations for obtaining fill material during both the construction and operation phases of the landfill.

# 1.3 Operation Plan Organization

The organization of the Operation Plan is described below:

- Section 2 describes personnel requirements, landfill entrance procedures, traffic routing, and facilities for the Class I landfill;
- Section 3 discusses landfill operations including basic landfilling procedures, waste handling, waste relocation, equipment, on-site roads, and general maintenance procedures for drainage swales;
- Section 4 discusses environmental controls including leachate containment and control, surface-water control, facility inspection, maintenance, monitoring, landfill active area controls, and record keeping;
- Section 5 describes the contingency plan for emergencies at the site;
- Section 6 describes the safety plan for the site;
- Section 7 discusses final closure of the JED facility; and
- Section 8 describes operation of the borrow area.

# 2 LANDFILL PERSONNEL AND FACILITIES

# 2.1 Personnel Requirements

# 2.1.1 Numbers and Types

The positions and number of personnel anticipated to be employed for each position are presented in Table 1. WCOC will have at least one trained operator at the landfill during active operations and at least one trained spotter at each working face. Spotter(s) will be located on heavy equipment spreading wastes while at the working face. All heavy equipment operators performing spotting duties while operating heavy equipment will be trained as an operator or spotter. The staffing levels presented in Table 1 provide for absences due to vacation, illness, holidays, or other reasons. Peak solid waste receipt periods, or other emergency conditions may require additional personnel and/or staff working overtime. These staffing levels are based on the assumption that work activities will generally take place 10 hours per day, 5 days per week and a half day on Saturday and Sunday.

If the daily volume at the landfill increases enough to require additional equipment, the staff will be increased as required to supply the personnel to operate and maintain the additional equipment. The minimum crew required to operate the landfill for receipt of waste is also presented in Table 1. In addition to the permanent staff, casual labor may be hired for area clean-up, ground maintenance, and other intermittent activities as required.

# 2.1.2 Employee Training

Employees of the landfill will receive initial and continued training in accordance with requirements of subsection 62-701.320(15), F.A.C., and other WCOC on-the-job training in the safe and environmentally secure operation of the landfill. In accordance with subparagraph 62-701.320(15)(b)1, F.A.C., the operators at the landfill will complete 24 hours of initial training and pass the examination as part of the initial training. Within three years of the initial training, and every three years thereafter, landfill operators will complete additional 16 hours of continued training. In accordance with paragraph 62-701.320(15)(c), F.A.C., the spotters at the landfill will complete 8 hours of initial training as described in this Operation Plan. Within three years of the initial training, and every three years thereafter, landfill spotters will complete additional 4 hours of continued training. All certified training (initial and continuing education) will be provided by a Florida Department of Environmental Protection (FDEP) approved, qualified, independent third party in accordance with Section 403.716, Florida Statutes (F.S.).

WCOC will maintain training records for current employees at the facility and will make all records available to FDEP upon request. The requirements of the training program will

also be documented in writing. Examples of subjects to be covered in the employee training program include the following:

- overview of this Operation Plan;
- review of permits and regulations for operators and other key personnel;
- general landfill safety procedures pertaining to work around solid waste, landfill gases, and leachate;
- instruction in the operation and maintenance of equipment, machinery, and systems which the employee must operate, service, or monitor during his/her daily job duties;
- instruction in emergency response procedures for landfill fires or explosions, leachate pumping system failure or leaks, or other emergency situations;
- instruction in emergency shutdown procedures; and
- appropriate procedures for spotters and equipment operators, scale masters and other key personnel including recognition of hazardous wastes and reporting procedures for discovery of unauthorized wastes.

A list of trained spotters and operators at the JED facility and a list and schedule of the classes offered to the public, which may be attended by the JED facility's operators and spotters, is presented in **Attachment A**.

# 2.2 <u>Landfill Entrance Procedures</u>

# 2.2.1 Hours and Days of Operation

Typical landfill hours for acceptance of waste are:

Monday through Friday: 4:00 am to 4:00 pm

Saturday: 6:00 am to 12:00 pm

Sunday: 6:00 am to 10:00 am

Construction, daily cell preparation, hauling/excavating, road building, leachate management, or all non-disposal waste acceptance can be performed both within and outside of the posted operating hours. The actual hours of operation will be posted at the main entrance to the landfill. The landfill may be closed on Sundays, Thanksgiving, Christmas, and New Year's Day.

# 2.2.2 Processing Customers

Upon entering the site, all landfill users entering the disposal area will be required to stop at the weigh station. The scale master will record the weight and type of waste for each waste load brought to the landfill. All waste loads will be visually inspected for hazardous or other unauthorized wastes in accordance with the waste inspection plan, which is presented in **Attachment B**. Unauthorized waste includes yard trash which cannot be disposed in a Class I facility. Customers will be directed to the City of St. Cloud, Osceola County, or other registered/permitted yard trash processing facility. A load-checking program will be used at the landfill to detect and discourage attempts to dispose of unauthorized wastes at the landfill. The load checking program consists of the following:

- Trained staff are required to examine at least three random loads each week. The
  selected waste hauling vehicles are to be directed to discharge their loads at a
  designated location within the landfill for a detailed inspection of the discharged
  material for any hazardous waste.
- If any regulated hazardous wastes are identified by the random load inspection or otherwise discovered to be improperly deposited at the landfill, the Landfill Site Management will promptly notify FDEP, and if known the person responsible for shipping the wastes to the landfill, and the generator of the wastes. The area where the hazardous wastes are found will be immediately cordoned off from public access and properly removed from the designated location/work face. If the generator or hauler cannot be identified, the landfill operator will assure the cleanup, transportation, and disposal of the waste at a permitted hazardous waste management facility.
- A record of information and observations gathered during each random waste load inspection will be maintained. This documentation will include: the date and time of inspection; load weight; names of the hauling firm and driver of the vehicle; vehicle license plate number; source of waste as indicated by the driver; and observations made by the inspector during the detailed inspection. The responsible inspector will sign each waste inspection record. The random waste load inspection documentation will be maintained at the landfill for a period of at least three years.

Vehicles will be directed to the appropriate disposal area by signs or other means. Verbal instructions will be given by facility personnel when necessary. The appropriate area depends on whether the waste is typical municipal solid waste, white goods, used tires, or waste that should be placed in a particular location for special handling.

#### 2.2.3 Public Use

Small, private vehicles will be directed to place their load in the appropriate disposal area by the scale master. Private vehicles will typically be directed to unload in two 20-yard roll-off containers located in the administrative area. However, private vehicles with a dump trailer will be directed to the landfill. These vehicles will be directed to unload in an area away from the commercial waste trucks and will be assisted to unload and return as quickly as practical.

# 2.2.4 Vehicle Inspection

A plan will be implemented by the Landfill Site Management to prevent the on-site disposal of unauthorized wastes. A copy of the Waste Inspection Plan prepared for the Class I landfill is presented in **Attachment B**. This plan will be implemented by the Landfill Site Management or designee to prevent the on-site disposal of unauthorized wastes.

The Landfill Site Management or designee (Inspector) will be in charge of inspecting waste vehicles arriving at the site. The Inspectors will receive training in unauthorized waste identification. The training provides the opportunity to improve the inspector's knowledge and ability to effectively screen incoming waste.

# 2.3 Traffic Routing

#### 2.3.1 Access Points/Signs

Access by all vehicles shall be via a single secured site entrance located on highway US 441. The entrance will allow for safe and orderly traffic flow into and out of the facility. The site entrance gate will be locked outside of operation hours.

Signs will be posted at the site entrance indicating the name of the facility, name of the operating authority, and hours and days of operation. In addition, a sign which clearly states "NO HAZARDOUS WASTES ACCEPTED" will be located at the entrance to the landfill. Traffic control and safety requirement signs will be located at and near the entrance to the facility as required.

#### 2.3.2 On-Site Traffic Flow

Once vehicles delivering wastes have been weighed, they will follow directions or signs posted along the haul road(s) to the current active work areas of the landfill, or designated offloading area for waste solidification and waste tire storage and processing operations. Trucks will then proceed to deposit their loads at the appropriate working face. Signs or

the scale master will direct small public vehicles to deposit their loads in the appropriate disposal area.

# 2.4 On-Site Structures

The site includes the following structures:

- office building/ticket office/weigh station
- scales
- maintenance shop
- storage area

# 2.5 <u>Communication Facilities</u>

The following communication facilities will be provided for routine communication and for use in emergencies at the site:

- cellular and/or conventional telephone in the office building; and
- on-site two-way radios.

#### 3 LANDFILL OPERATIONS

# 3.1 Basic Landfilling Procedures

This section describes the procedures that constitute the daily landfill operations, the sequence of landfilling, working face practices, and control of the first and subsequent lifts. The landfill will be operated in accordance with these procedures and filled in the general sequence as indicated on the Permit Drawings.

# 3.1.1 Method of Operations

Landfilling areas will generally progress from north to south and from west to east. When a cell is opened, waste lifts will be placed in areas that do not have a rain cover to a depth of 10 to 15 feet to reduce leachate generation prior to placement to higher elevations in a cell.

Controlling truck routes and properly spotting loads will facilitate the spreading, compaction, and covering of refuse. During construction of the first lift, trucks will be positioned on a lift of previously compacted waste adjacent to the first lift being placed. In subsequent lifts, unloading at the toe of the working face and pushing uphill may be the preferred method. Lateral confinement or small work faces will be maintained to avoid wasting soil cover material. Temporary barricades or flags may be used as daily width markers for guiding equipment operators and for traffic control.

Vehicles transporting refuse and cover material to the working face will be routed over previously filled areas, whenever possible, for additional compaction of refuse and soil. Vehicles will not be routed over areas of the final cover system unless on a road specifically designed for hauling waste. Disposal vehicles will not be routed over a lined area before a lift of waste has been placed, in order to prevent damage to the liner.

Signs will also be posted in the operational areas if and when required. These signs will direct traffic, identify buildings, and specify types of material to be deposited in particular areas, including the waste solidification and waste tire storage areas. Safety signs will also be posted to identify certain safety requirements such as no smoking, speed limits, and stop signs.

The refuse may consist of household wastes, non-hazardous industrial wastes and sludges, commercial wastes, construction and demolition (C&D) debris, and other similar materials, as allowed by regulations for Class I landfills. These readily compactable wastes lend themselves to the typical operations described in Sections 3.1.2 and 3.1.3. Any new, non-routine solid waste stream will be evaluated (i.e., waste profile) before it is received at the

landfill to ensure it is allowed for disposal as a Class I waste and, if not prohibited, to ensure it is then properly managed at the landfill.

# 3.1.2 Working Face Practices

# 3.1.2.1 Start-Up and First Lift

To assure protection of the landfill liner system, no disposal vehicles will be operated directly on the liner protective cover. Soil platforms or similar protective measures will be placed adjacent to the working face to keep vehicles off the liner protective cover. Landfill personnel will be positioned at the working face for the start-up of each new area to direct vehicles to their unloading points.

The first lift of waste on the liner protective cover will be placed with great care, using special methods to protect the liner from damage. The first lift of waste will be a minimum of 4 feet in compacted thickness and consist of select wastes containing no large rigid objects that may damage the liner or leachate collection system. Equipment will not be allowed on the liner protective cover and equipment will not spread waste in a manner that displaces the liner protective cover soil. Landfill personnel will closely monitor the placement, compaction, and covering of the first layer of waste. Landfill personnel will maintain grade control and inspect the filling techniques. Inadvertent damage or suspected damage to the liner system will be reported to the Landfill Site Management and restored prior to filling in the damaged area.

To protect the liner system, the bulldozer will normally be used as the primary spreading and compacting machine for the first lift. The compactor will only be operated on top of the waste and not on the landfill base or on the waste sideslopes. The equipment operators will also make sure that no bulky waste or other material, which could damage the liner system is placed within the first lift.

To reduce the quantity of leachate generated during the initial stages of waste placement, plastic sheeting may be deployed to collect storm water generated in those portions of the cell where waste materials have not yet been placed. The area of the cell covered by the rain cover will be modified as necessary to accommodate waste placement. Temporary berms will be constructed down gradient of the working face such that impacted storm water from the operations area and the deposited waste will not be able to co-mingle with the un-impacted storm water. The temporary sheeting will be secured with sand bags, tires, or other equivalent ballasting that will not puncture the temporary sheeting. The temporary sheeting will extend from the top of the intercell and side slope berms across the entire width of the cell.

Un-impacted storm water will be collected in a temporary sump, constructed on top of the liner protective layer and pumped to the existing storm water management system.

Dedicated storm water pumps, (i.e., electric submersibles or gasoline powered trash pumps) will pump the storm water from the top of the plastic sheeting. The un-impacted storm water will be pumped to the nearest storm water down chute such that the pumped water will flow out of the down chute onto the rip rap placed at the end of the outlet structure. The pump head of this type pump is placed in the low point of the cell and is capable of drawing down liquid to within approximately 12 inches of the containment bottom.

The rain cover will be removed prior to placement of waste in cell areas such that waste is placed in direct contact with the liner protective layer.

# 3.1.2.2 Subsequent Lifts

After the first lift is properly in place, normal operating procedures will be used for the second lift and all subsequent lifts. Trucks and compactors are permitted to operate on these lifts. Bulky wastes delivered to the facility and any stockpiled bulky wastes received during construction of the first lift will be placed in subsequent lifts. The daily operating procedures including routing of traffic, placement, spreading and compaction of refuse, and application of initial and/or intermediate cover will be followed for the subsequent lifts of waste. Soil erosion control and site maintenance tasks will be implemented throughout the development of all lifts. Once the final landfill elevations have been reached over a suitably sized area, final cover may be applied to the landfill during the next construction season and vegetated during the customary planting season. Areas at final grade and interior slopes with intermediate cover may be covered with a temporary exposed geomembrane cover for erosion control purposes until such time final closure occurs or waste filling operations resume.

At the end of each working day, initial cover material (e.g., soil or alternate material) will be applied. When soil is placed as initial cover, an excavator, loader and truck will be used to load and haul soil from the stockpile area to the working face where it will be temporarily stockpiled or spread directly over the waste. Alternate initial cover materials are listed in Section 3.1.4.2, and will be used as described in that Section. Intermediate cover will be applied on areas that will be exposed for more than 180 days (i.e., outside side slopes and the top of the final lift or portions of other lifts not soon to be covered by additional refuse.) As previously noted, a temporary exposed geomembrane cover may be installed over the intermediate cover for erosion control.

Material from on-site stockpile or borrow areas will be used to supply initial and intermediate soil cover requirements. To conserve soils and landfill space, the initial and intermediate cover will be scraped back immediately before placement of additional solid waste on top of the lift, and then reused as cover material if appropriate, or will be incorporated into the working face. Areas with initial, intermediate, and temporary exposed geomembrane cover will be graded to drain away from the active work area.

# 3.1.3 Filling Procedures

After the first lift, waste materials will be placed in approximately 2-ft thick horizontal layers when possible and compacted to approximately 1-ft thickness or as thin a layer as practical before the next lift is applied. Lift depths will typically not exceed 10-ft in thickness and the working face slope will not be greater than 3-ft horizontal to 1-ft foot vertical rise. However, a lift thickness of 10 to 15-ft may be necessary at times during days of higher waste acceptance rates and fill transition at outer slopes and change in direction. The need for an increased lift thickness is highly dependent on the fill configuration on that particular day and actual waste acceptance rate. An increased lift thickness is allowed by paragraph 62-701.500(7)(c), F.A.C. When operating with an increased lift thickness, WCOC will maintain adequate widths and spacing of operations as further described below to ensure safe operating practices.

The refuse cell is the basic building block of a landfill. It is composed of multiple compacted layers of waste and enclosed by cover material (i.e., initial, intermediate and/or final cover). Basic instructions for constructing the refuse cell are outlined below.

# 3.1.3.1 Width of Working Face

The working face is the portion of the uncompleted cell on which additional waste is spread and compacted. To maintain sanitary operation, the working face will be kept as narrow as possible. By keeping the working face narrow, equipment movement, cover material requirements, and the area of exposed waste is minimized. In order to facilitate proper unloading and waste placement operations, two working faces may be required from time to time.

The optimal daily working face width will vary depending on the number of vehicles bringing waste to the site. The working face will be wide enough to prevent a large backlog of trucks.

#### 3.1.3.2 Unloading

When unloading waste from top of the refuse cell, the waste will be discharged as close to the edge of the active working face as safe operations permit and pushed down slope. For safety reasons, a minimum 8 to 10 ft separation will be maintained between the refuse trucks and the landfill equipment.

When unloading waste from the bottom of the refuse cell, the waste will be discharged approximately 10 ft from the toe of the working face and pushed up the slope. Truck and landfill equipment separation, as discussed above, will be maintained. In order to prevent loads of waste from being discharged too far away from the toe, refuse trucks can be

backed toward the toe, following a path created by the equipment pushing refuse into the working face.

# 3.1.3.3 Pushing, Spreading, and Compacting

Proper refuse cell construction involves pushing, spreading, and compacting the waste. These functions will be accomplished with a bulldozer and/or a compactor.

Pushing the waste is the action of moving the waste from the discharge location into the working face. This function will be accomplished with a bulldozer and/or compactor.

Spreading of the waste can be done by either a bulldozer or compactor. The purpose of the spreading action is to distribute the waste over the working face in a thin layer (approximately 2 ft thick). High in-place compacted unit weight of the waste is achieved by compacting in thin layers (i.e., 2 ft thick).

Good compaction is achieved by operating the landfill compactor up and down the working face after the refuse has been spread into a thin layer. Proper compaction of the waste will extend landfill life, while reducing litter and vector problems. To maximize compaction of the waste, the working face and inside temporary slopes will not exceed a maximum slope of 3H:1V. The Landfill Site Management will periodically verify the compaction procedures and make corrections as necessary.

# 3.1.3.4 Daily Clean-Up

The area receiving wastes will be policed daily for loose waste and litter. Such waste, as well as litter along the litter fences, will be removed. The litter may be stored in trash bags until it can be deposited in the landfill.

#### **3.1.4** Cover

#### 3.1.4.1 Stockpiling

Cover soil stockpile locations, if needed, will change throughout the life of the landfill depending on site conditions and the location of the active working face. Landfill equipment will begin pushing or spreading the cover over the active cell area when and where it has reached its limit for the day.

#### 3.1.4.2 Application and Phasing of Cover Materials

A 6-in. thick soil initial cover will be placed on top of all exposed waste on the working face at the end of each day's operation unless additional waste is to be deposited on the working face within 18 hours. The initial cover may also consist of alternative materials provided their use meets the criteria of subsection 62-701.200(53), F.A.C. Alternate initial cover materials that may be used at this landfill include Posi-Shell, tarps, processed tire

chips, auto shredder residuals, mulch mixed with soil at a maximum 50/50 ratio, petroleum contaminated soils, the additional materials authorized by rule in paragraph 62-701.500(7)(e), F.A.C., and the alternate initial cover products already approved by the FDEP that are listed at: <a href="https://floridadep.gov/waste/permitting-compliance-assistance/content/solid-waste-guidance-memos-documents">https://floridadep.gov/waste/permitting-compliance-assistance/content/solid-waste-guidance-memos-documents</a>. All alternate initial covers will be used in a way to ensure they meet the criteria of subsection 62-701.200(53), F.A.C. Posi-Shell will be prepared and used in accordance with the manufacturer's instructions and will be applied to the waste working face with appropriate equipment such as a commercial hydro-seeding unit.

A 12-in. thick intermediate earth cover will be placed over the initial cover within 7 days of completion of an area if no additional solid waste will be deposited within 180 days. As noted in Section 3.1.2, a temporary exposed geomembrane cover may be placed over the intermediate cover materials to facilitate erosion control.

Final cover will be placed over the areas of the landfill that have reached final design elevations. Final cover will be placed within 180 days of reaching the final design elevations. Temporary exposed geomembrane covers will be removed, if present, prior to placement of final cover. The final cover system will be as described in Section 7 of this Operation Plan. Vegetation will be maintained over the final cover areas throughout the life of the landfill and the post closure care period. Maintenance of the final cover swales, and access roads will also be performed throughout the life of the landfill and the post closure care period.

# 3.2 **Equipment**

#### 3.2.1 Primary Equipment

Based on the available range of handling capacities and the initial projected waste receipts, the allocation of heavy, primary equipment presented in Table 2 will be sufficient to handle the wastes received at the landfill. The primary functions of heavy landfill equipment are spreading and compacting solid waste, and excavating, hauling, and spreading cover material. Equipment similarities allow different equipment to perform functions as necessary. For example, when a compactor breaks down, a bulldozer can perform the compaction operation.

# 3.2.2 Back-Up Equipment

The equipment selection guide indicated in Table 2 will be adequate even if one of the pieces of equipment is temporarily out of service. If a piece of equipment is out of service for an extended period or if additional equipment is required on a temporary basis, this

equipment is available for rental from several heavy equipment rental companies listed in Table 3.

# 3.2.3 Support Equipment

In addition to the heavy equipment used for operating and maintaining the landfill, other support equipment may be used to perform work not essential to the operations. This equipment will be present at the site most of the time, but some may be off-site, temporarily out of service, or rented for a specific occasion.

One 3,000-gallon or larger water truck will remain on site at all times and will be used for dust control and fire protection. The water truck will normally be positioned close to the working face for fire protection and will be equipped with spray bars so it can be used for dust control.

A utility tractor will be used to perform site maintenance activities. It will be fitted with attachments for mowing grassed areas. A backhoe or small excavator will assist the small dozer in maintaining drainage courses and ditches and for other site maintenance activities.

Pumps will be used for filling the portable water storage tank. These pumps will also be used to dewater any ponded water that forms in low areas around the site, including roads and lined landfill areas not in use.

# 3.2.4 Equipment Care

Routine preventive maintenance will be performed near the active landfilling area or at the maintenance facility to minimize equipment downtime and increase equipment service life.

Preventive maintenance varies with each piece of equipment. Therefore, the operation and maintenance (owner's) manual for each should be consulted. However, three applicable maintenance activities, which will be implemented at the site are:

- establish a routine equipment inspection program
- lubricate according to manufacturer's recommendations
- keep maintenance records

# 3.3 Roads

#### 3.3.1 Road Construction

The main access road from the site entrance area to the scale house will be paved or covered with suitable aggregate material. Haul roads from the scale house to the active work area in the landfill will be improved, all weather, rock/recycled concrete surfaced or

paved. A perimeter maintenance road will provide all weather access to leachate management systems, groundwater monitoring wells, landfill gas monitoring wells, and storm water management structures. The perimeter maintenance road and roads in the active work area will be surfaced with suitable aggregate material.

#### 3.3.2 Maintenance of Roads

# 3.3.2.1 Filling of Potholes

Potholes will be filled with materials compatible with the road construction material. Potholes will be filled on a routine basis so that they are not allowed to remain open for extended periods. Before placing patches in holes, all loose material will be removed from the hole. New material will then be placed in the hole and compacted so that it will be approximately as dense as the materials originally used in the road.

# *3.3.2.2 Grading*

As unpaved, all-weather roads become uneven due to traffic-caused rutting or displacement of stone, fresh rock or recycled crushed concrete will be applied to the surface and smoothed to an evenly sloped grade to promote drainage.

# 3.3.2.3 Restoring Settlement

When all-weather roads are built on fill areas, settlement of the filled area may cause cracks to appear in a road or cause the slope of a road to change. Cracks will be filled with material that is compatible with the roadbed. Areas of a sloped road, where the slope has changed drastically, will be built up with material compatible with the roadway and constructed utilizing acceptable construction methods until the desired elevation is achieved or the road section is rebuilt.

# 3.3.2.4 Cleaning of Public Access Roads

Proper operation of the landfill will result in little or no debris being found on public roads. The public roads adjacent to the site entrance area will be inspected daily. If debris from the wheels of vehicles departing the landfill reaches the public access road at the entrance to the landfill, that road will be cleaned to a distance of 0.25 mi or as required in both directions, if necessary, from the entry point onto the road.

# 3.3.2.5 Removal of Materials from Landfill Roadways

Any significant accumulation of dirt, brush, and other debris will be removed from the landfill roadways. Dirt left on asphalt roadbeds may cause dust problems during dry weather or mud problems during wet weather. A program of road cleaning will be implemented to prevent any buildup. Unpaved roads will be watered as needed to minimize dust.

# 3.3.2.6 Maintenance of Drainage Swales

Drainage swales along road beds will be kept free of obstructions. During the wet weather seasons, inspection of all drainage ditches and structures will be made in accordance with the site Stormwater Pollution Prevention Plan (SWPPP), and debris removed as required.

#### 3.4 **Drainage Features**

- Inspections will be performed in accordance with the site SWPPP (outlined in Section 4.2.3).
- Channels, Pipes, and Inlet Structures: Drainage structures will be cleaned of debris
  as soon as practical after problems are identified to prevent ponding. When unlined
  channels silt up, routine cleaning will be performed to restore the original capacity
  of the channels.
- Repair of Structures: Damaged structures will be permanently repaired during dry weather periods. During rainy periods, temporary repairs may be made to prevent further damage to the structure or erosion of soil.
- Sediment Barriers: Sediment barriers will be visually inspected periodically for damage, and to determine if sediment has accumulated behind them. Sediment will not be allowed to accumulate to a height exceeding half that of the barrier. Barriers will be replaced when visibly damaged. Barrier footings will also be inspected to ensure that drainage is not flowing beneath the barrier unless designed to do so.

# 3.5 Salvaging/Recycling

No scavenging will be permitted at the landfill. Waste tires will be accepted, temporarily stored and transported to a licensed tire processor. Processed tires may be directly disposed in the landfill or used as initial/daily cover. Other recycling uses of processed tires may be proposed as markets are developed. Waste tire and processing operations are described in the Waste Tire Storage and Processing Plan provided in **Attachment C**. If the volume of other recyclable goods is sufficient, as determined by the Landfill Site Management, those items may be separated from the waste which is to be disposed. Landfill gas is planned to be converted to electric energy as described in the Landfill Gas to Energy Operation Plan provided in **Attachment D**.

#### 4 ENVIRONMENTAL CONTROLS

This section presents the basic components of the environmental controls at the JED Class I landfill. The major components of this section are the Facility Inspection Plan, Facility Maintenance Plan, and the Facility Monitoring Plan. In this section, a discussion of each of these components is presented, including a discussion of groundwater and surface-water protection controls, leachate collection system (LCS), and surface water controls, where appropriate. The discussion also includes general facility controls, including initial, intermediate and final cover, temporary exposed geomembrane cover, and access roads.

#### 4.1 Environmental Control Systems

The purpose and function of each of the major environmental control systems are described below. Specific construction and design details are presented in the construction documents, the closure plan, post-closure plan, and the design report with attached plans.

#### 4.1.1 Leachate Containment and Control

The Class I landfill is equipped with a double-composite liner system, which directs any liquid entering the landfill that may have contacted refuse to an LCS. The LCS drains liquid collected on the primary liner into a sump. Leachate in the sump is pumped into on-site leachate storage ponds and trucked to the Aqua Clean Environmental Company) or other permitted WWTPs for disposal or recirculated back into the lined landfill area in accordance with the procedures described in the Operation Plan for Leachate Recirculation provided in **Attachment E** or sent to the onsite leachate evaporation system as discussed in **Attachment H**. The on-site water truck will be used to recirculate leachate under the Surface Leachate Application method described in **Attachment E**. After each use of the water truck for leachate recirculation, one load of clean fresh water will be sprayed within the lined disposal boundary prior to using the water truck for dust control purposes outside of the disposal boundary limits. Quantities of leachate collected by the LCS will be recorded in gallons per day and maintained as part of the landfill operating record.

A recording rain gauge will be installed, operated, and maintained to record precipitation at the landfill. Precipitation records will be maintained as part of the landfill operating record and used to compare with leachate generation rates.

#### 4.1.2 Surface Water Controls

The surface-water management system for the JED facility consists of a system of drainage swales to move storm water to either permanent dry retention basins or interim dry retention basin, depending on the stage of landfill construction. All dry retention basins are surrounded by an earth berm designed to contain all runoff from a 100-year, 24-hour storm event.

Where runoff must pass through a roadway, appropriately sized culverts will be installed.

# 4.2 Facility Inspection Plan

# **4.2.1** Leachate Collection System (LCS)

The LCS will be water pressure cleaned or inspected using a video camera after construction but prior to placement of any waste. The pump(s) will be tested in the sump to assure that the system operates properly. Deficiencies will be repaired prior to initial deposition of waste.

The LCS includes manholes, pumps, a leachate wet well and two force mains, valves and cleanouts. The LCS pumps will be removed and inspected every 2 years. This 2-year inspection will consist of pressure testing of the pump. Pumps located in active areas, or areas without final cover, will be inspected on a monthly basis to confirm normal operation. Additional inspection, preventative maintenance, and checking of the electrical components will be performed in a manner and frequency in accordance with manufacturer's recommendations. The leachate transmission manholes will be inspected weekly for accumulation of leachate in the manhole and to verify integrity of the force main. Each leachate force main will be inspected and cleaned (if needed) every 5 years.

# 4.2.2 Leachate Storage Facility

The exposed exterior of the polyethylene leachate storage ponds will be inspected weekly for leaks, deterioration, and maintenance deficiencies. The leachate level and overfill control equipment will also be inspected weekly to ensure it is in good working order. The leachate levels in the ponds are inspected and recorded on a daily basis.

If inspection reveals a storage container or equipment deficiency, leak, or any other deficiency that could result in failure of the storage system to contain the leachate, remedial measures will be taken immediately to eliminate the leak or correct the deficiency. Inspection reports will be maintained and made available to FDEP upon request for the lifetime of the leachate storage facility.

#### **4.2.3** Surface Water Control System

Surface-water culverts may contain landfill gas. Prior to accessing piping, protective measures will be taken to avoid explosion(s), fire(s), and asphyxiation(s).

Drainage swales, inlets, structures, and the surface-water management areas will be visually inspected in accordance with the site SWPPP. The frequency of dry inspections may be modified as appropriate based on progressive experience with the landfill drainage system, however, in no case will inspections be less frequent than quarterly. Regardless of

the inspection frequency, the system will be inspected following each 25-year, 24-hour storm event (i.e., 9 inches of rain in 24-hour period) or greater storm event.

Drainage swales, inlets, and structures will be cleared of obstructing debris as soon as practical after a problem is identified. If channels become filled with an accumulation of debris or soil, cleaning may be required to restore original flow capacity.

Erosion control measures will be employed to correct any erosion that exposes waste or causes malfunction of the storm water management system. The control measures to repair erosion will be implemented within three days of occurrence. If the erosion cannot be corrected within seven days of the occurrence, a proposed correction schedule will be provided to FDEP.

# 4.2.4 Landfill Cover System

Areas that have received intermediate or final cover will be visually inspected periodically for signs of erosion, cracks and depressions due to settlement, and leachate seeps. Areas where waste or geosynthetics have been exposed by erosion will be filled and regraded to minimize any subsequent erosion. Significant depressions (1 ft or more) will be filled with soil, compacted, and regraded to promote positive drainage. If leachate seeps appear in the uncapped area of the landfill, the seep area will be excavated and backfilled with highly permeable material to promote seepage through the landfill. The intermediate cover will be reworked to seal the area.

Areas that have received temporary exposed geomembrane cover will be visually inspected periodically for signs of damage (e.g., puncture, tearing, etc.) and depressions due to settlement. Geomembrane damage will be repaired if it is resulting in significant undercutting (1 ft or more) of the geomembrane cover. Significant depressions (1 ft or more) that result in water ponding on top of the temporary exposed geomembrane cover will be remediated as follows: i) the geomembrane will be removed from the depression area, ii) the depression will be backfilled with compacted soil and regraded to promote positive drainage, and iii) the temporary exposed geomembrane cover will be placed back over the regraded area.

# 4.2.5 Facility Inspection Schedule

Daily Levels in leachate storage ponds

Weekly Exterior of HDPE leachate storage containers and overfill control

equipment

Monthly Leachate collection pumps

(Visual) Surface-water management system

Cover in completed areas

Leachate force main

Quarterly Surface-water control system (or after a 25-year, 24-hour storm event)

Annually Surface-water control system pipes and structures

Topographic survey of landfill

Bi-Annually LCS pumps and pipelines

(every 2 years) Leachate collection and detection flow meters, valves, and risers

#### **4.3** Facility Maintenance Plan

In conjunction with the inspection plan, a regular schedule of maintenance will be prepared and implemented. This section refers specifically to the maintenance of the environmental controls installed at the landfill. It does not include the regularly scheduled maintenance of landfill roads or equipment such as vehicles, scales, or buildings. Maintenance requirements in this section refer primarily to the mechanical equipment associated with environmental controls. In addition, each piece of equipment will be inspected and maintained in accordance with all manufacturer's recommendations.

# **4.3.1** Leachate Collection System

The electrical controls, pumps, flow meters, valves, and couplings will be maintained on at least a bi-annual basis (i.e., every two years). In addition, parts that tend to wear out on a regular basis, including bearings on pumps, seals, and gaskets, will be replaced during regular maintenance. After replacing maintained parts, the equipment will be tested to assure proper performance.

# 4.3.2 Surface-Water Control System

The surface-water control system does not include mechanical systems that require regular maintenance, however, the system is to be inspected in accordance with the site SWPPP. The swales, drainage structures, inlets, and pipelines will be repaired and maintained as soon as practical following identification of any damage or deficiencies. This includes repair of lined and unlined ditches in the active landfilling areas, on intermediate and final cover and diversion ditches around the landfill.

#### **4.3.3** Final Cover Maintenance

Maintenance of the final cover includes all the components of the cap, i.e., the geomembrane, drainage geocomposite, protective soil layer and vegetation. The periodic inspections will help in assessing the final cover condition to verify the integrity of the cap

(e.g., check for cracking of protective cover layer due to differential settlement or erosion and exposure of cover geomembrane/geocomposite), and the condition of the vegetation.

Areas of ponding or substantial differential settlement (1ft or more) will be checked to determine the cause. If a significant problem with the cover, vegetation, perimeter berms, erosion, or drainage structures is identified, work orders will be issued to correct the problems. Repair work shall be initiated as quickly as possible.

The timing of the repairs will be dependent on the nature of the repair. Minor filling to eliminate ponding, and the reseeding and fertilizing disturbed or problem areas will be accomplished with little delay. Major repairs, such as extreme erosion, significant local instability of slopes, or substantial settlement, might require geotechnical evaluation and design prior to implementing final repairs. In some cases, the need for analysis and design of the severely damaged areas will delay repair activities.

If repairs are necessary to the cover system swales, inlets, or downdrains to correct the runoff containment system deficiencies, the repairs will be undertaken prior to start of the wet weather season.

Repair of damages to the cover system resulting from erosion and differential settlement may include locally removing geosynthetics and backfilling depressions beneath the geomembrane, repairing geosynthetics, backfilling soil layers, and revegetating disturbed areas. Additional drainage facilities may be provided to prevent future erosion.

# 4.4 Water Ouality Monitoring Plan

The groundwater and surface water quality monitoring plan for the JED facility is presented in Appendix E of the 2011 Renewal Permit Application.

#### 4.5 Landfill Gas Monitoring Plan

The landfill gas monitoring plan proposed for the JED facility will allow early detection of the lateral migration of landfill gas and verification of the landfill gas management system performance in accordance with the requirements of subsection 62-701.530(1), F.A.C. The following types of landfill gas monitoring will be performed at the site: (i) monitoring for landfill gas in on-site buildings; (ii) monitoring for landfill gas migration along the perimeter berm; and (iii) monitoring at the property boundary for objectionable odors. The following subsections provide a description of the gas monitoring that will be performed at the JED facility.

# 4.5.1 Monitoring of On-Site Buildings

The on-site buildings will be located in the entrance area of the landfill. All buildings located within 500 feet of the waste limits on the property will be routinely monitored for methane. Continuous monitoring devices used within on-site buildings will be located in work areas, near any penetrations or cracks in building foundation, or at points where methane might enter the building.

If methane is detected at a concentration greater than 25 percent of the lower explosive limit (LEL) in any on-site building, WCOC will perform the activities described in Section 4.5.4.

# 4.5.2 Monitoring for Landfill Gas Along Property Boundary

Gas monitoring probes along the property boundary (or the storm water berm where property boundary is far away from the waste limits) will be used to detect lateral migration of landfill gases. The gas monitoring probes located around the perimeter of the site will be monitored quarterly for methane. Should the results of the quarterly monitoring indicate lateral migration of landfill gases, WCOC will install additional gas monitoring probes, as needed, in the area(s) of concern and/or perform additional monitoring. If methane is detected at a concentration greater than the LEL in the gas monitoring probes at the property boundary, WCOC will perform the activities described in Section 4.5.4 below.

# 4.5.3 Monitoring for Objectionable Odors at the Property Boundary

WCOC's on-site personnel will operate the facility to control objectionable odors and will perform monitoring for objectionable odors at the property boundary on a regular basis. It should be noted that no off-site occupied structures currently exist near the property boundary. Upon notification from the FDEP that objectionable odors have been confirmed beyond the property boundary, WCOC will:

- Immediately take steps to reduce the objectionable odors. Such steps may include applying or increasing initial cover, reducing the size of the working face, and ceasing operations in the areas where odors have been detected;
- Submit to FDEP for approval an odor remediation plan for the gas or odor releases. The plan shall describe the nature and extent of the problem and the proposed long-term remedy. The remedy shall be initiated within 30 days of approval;
- Implement a routine odor monitoring program to determine the timing and extent of any off-site odors, and to evaluate the effectiveness of the odor remediation plan.

# 4.5.4 Detecting Exceedances of the Regulations

Should the results of the gas monitoring indicate that the requirements of subsection 62-70.530(1), F.A.C., have been exceeded at the facility, WCOC will:

- immediately take all necessary steps to ensure protection of human health and notify the FDEP;
- within 7 days of an observed exceedance, WCOC will submit to the FDEP for approval, a plan to remediate the landfill gas migration; and
- within 60 days of an observed exceedance, WCOC will complete the remediation, unless otherwise directed by FDEP.

# 4.6 <u>Landfill Active Area Controls</u>

#### 4.6.1 Litter Control

Maintaining proper litter control is essential to the operation of a landfill. When working in areas below natural grade, litter is less likely to escape than when working above natural grade. Litter control procedures for the landfill are discussed below.

# 4.6.1.1 Prevention of Litter on the Working Face

Litter will be minimized as follows:

- Following proper techniques at the working face may prevent a considerable amount of potential litter by reducing the amount of refuse exposed to the wind. Where possible, the exposed face of the cell will be oriented into the wind. This will cause the wind to blow any loose litter back into the working face and helps keep waste away from the undercarriages of unloading vehicles, which may track the waste along the public roadway as they exit the facility. The compacted waste already on the face helps trap litter.
- When top discharging, refuse will be placed as usual and spread downward. When possible, the exposed face of the fill will be oriented away from the wind for the same reasons bottom discharging is oriented into the wind.
- Compacted waste will be covered as soon as practical to minimize blowing litter.

# 4.6.1.2 Control of Litter with Litter Fences

Litter that escapes from the working face of the fill area may be controlled by litter fences. Movable/permanent fences may be positioned near the working face as wind and fill

operations change. Permanent litter fences may also be placed around the perimeter of the fill areas for additional litter control.

#### **4.6.2** Buffer Maintenance

Litter may occur even with proper litter controls. The following clean-up and maintenance procedures will be followed on a routine basis to maintain the buffer areas:

- Litter clean-up from along fences and buffer vegetation: Litter will be removed from and along litter fences and vegetation as necessary. Litter will not be allowed to accumulate in buffer vegetation.
- Clean-up along on-site roads and buffer areas: Litter occurring along on-site roads and in buffers will not be allowed to accumulate. This litter will be cleaned up as necessary.
- Clean-up at entrance area and entrance road: The site entrance and the road leading to the entrance (¼ mile each direction) will be inspected daily. These locations will be cleared of litter as necessary.
- Vegetation will be maintained and supplemented as necessary in order to provide an adequate visual screen.

#### 4.6.3 Dust Control

Dust control will be practiced during operation of the landfill by the application of water sprays from a water tank truck. The frequency of application of water for dust control will depend on site conditions and specific operation being performed. When necessary, water will be sprinkled on all heavily used roads. The main access road will be regularly sprayed to control dust when required.

#### 4.6.4 Vector Control

Vectors, animals, or insects will be minimized. Maintaining the working face as narrow as possible, providing initial cover on exposed areas, and eliminating water ponding are the primary safeguards against vector problems. Well-compacted wastes and cover material effectively prevent vectors emerging from or burrowing into wastes. Inspections of the waste tire storage area will be performed as necessary to monitor the area for the presence of rodents and mosquitoes.

If problems with rodents or insects occur, monitoring and surveys for vectors will be conducted to verify the effectiveness or identify and implement improved vector control practices, including the use of traps and insecticides.

#### 4.6.5 Noise Control

All equipment powered by internal combustion engines will have mufflers installed and maintained in good repair. Screening berms will also be used, when possible, to deflect sound upward.

#### 4.6.6 Recordkeeping

An operating record will be maintained at the site including all records, reports, analytical results, demonstrations and notifications; any construction, operation, and closure permits, including all modifications to those permits, issued by the FDEP, along with the engineering drawings and supporting information; as well as training verifications. This record will be kept with the operation plan at or near the landfill facility, or in an alternate location designated in the operating permit which is readily accessible to landfill operators. The operating record will be available for inspection at reasonable times by the FDEP and maintained for the design period of the landfill.

As part of the operating record, waste records will be maintained. These waste records will indicate the amount of each type of solid waste received each day, and the reporting and record keeping procedures outlined in the Waste Tire Storage and Processing Plan (Attachment C), LFGTE Facility Operation Plan (Attachment D), and Waste Solidification Operation Plan (Attachment F). Waste reports, summarizing the waste records will be compiled monthly and copies will be provided to FDEP annually. The waste records will be kept with the operation plan at the landfill and will be available for inspection at reasonable times by the FDEP. Weight receipt records will be kept for a minimum of five years.

The operating record will also include the information and observations resulting from each random inspection of a waste load conducted as part of the load checking program as described previously in Section 2.2.2.

The operating record will also include:

- the quantities of leachate collected by the primary leachate collection and removal system, and the secondary leachate detection and removal system, in gallons per day; and
- a record of the daily precipitation at the landfill based on the rain gauge installed, operated and maintained at the landfill.

This data will be used to calculate the monthly leachate generation rates expressed as a percentage of the monthly precipitation.

In addition, the operating record will also include the following:

- records of all information used to develop or support the permit applications and any supplemental information required;
- records of all monthly information, including calibration and maintenance records, and water quality records; and
- an annual estimate of the remaining life and capacity in cubic yards of the existing, constructed landfill and remaining life and capacity of other permitted areas not yet constructed. This estimate will be reported annually to FDEP.

The operating records will be maintained at the landfill throughout the design life of the landfill. Records that are more than five years old which are required to be retained may be archived, provided that the landfill operator can retrieve them for inspection within seven days.

#### 5 EMERGENCY CONTINGENCY PLAN

#### 5.1 Introduction

This section identifies a set of unplanned circumstances that may occur at the landfill. If handled correctly, the damage or impacts from these problems can be minimized. This section presents procedures to follow for dealing with problems as they occur. Operating personnel will become familiar with the procedures in order to prevent environmental contamination or damage to landfill facilities.

The entrance to the facility allows emergency vehicles immediate access to the landfill by police, fire, and ambulance.

**Attachment G** presents a list of individuals responsible for implementation of this Emergency Contingency Plan, and facility and emergency response agencies to contact. This list will be posted near all telephones on-site to provide "ready" access to emergency response agencies.

Within 24-hours, FDEP, Central District, shall be notified of any fire or other emergency which poses an unanticipated threat to the public health or the environment. The notification can be made by telephone or e-mail at the contact information listed in **Attachment G**.

This plan is organized by subsection and contains specific plans to address each type of occurrence listed below:

- fire
- accident or injury
- release of contamination to environment
- hazardous waste
- uncooperative customers
- inclement weather
- problems with the leachate collection and leachate removal systems

# 5.2 Fire Control Plan

The procedures below will be followed in the event of a fire at the facility. Additional fire prevention measures for the waste tire storage and processing operations are presented in Section 6 of the Waste Tire Storage and Processing Plan provided in **Attachment C**.

#### **5.2.1** When Fire Occurs

The following procedures will be followed in the event of a fire at the facility:

- extinguish small fires with fire extinguisher or smother with soil do not remain near large fires or explosive materials;
- determine location, extent, type, and, if possible, cause of fire or explosion;
- notify on-site personnel and implement safety and fire control procedures;
- if the fire is located at the waste tire storage and processing area, determine if there is a possibility of discharge of oily materials generated by burning tires. If so, implement measures to contain the oily materials within the storage area (i.e., blocking the storm water discharge pipe at the storage pad and installing absorbent sock type materials;
- notify Landfill Site Management if the fire cannot be immediately controlled;
- notify fire department if necessary. Clearly state:
  - location of landfill,
  - location of fire or explosion in landfill,
  - extent of fire or explosion,
  - type of fire or explosion,
  - actions now being taken, and
  - injuries;
- notify rescue squad, if necessary;
- notify health care facility, if necessary;
- notify sheriff, if necessary; and
- notify cleanup company for oily material generated at waste tire storage area, if necessary.

#### 5.2.2 "Hot Load" Procedures

In the unlikely event that a "hot load" is not identified before entrance into the facility, the following procedures are implemented:

- the truck carrying the "hot load" is to be directed to discharge the load in the landfill but away from the working face and any exposed liner;
- the load is to be placed on top of intermediate cover which will provide sufficient protection between the "hot load" and the underlying waste;

- soil will be spread over the load to smother the "hot load"; and
- the "hot load" will be monitored until there is no evidence of smoldering or high temperatures.

At the end of the day or when appropriate, the load will be worked into the active working face. Areas where "hot loads" are extinguished varies depending on the location of the working face, but will always be away from the working face and any exposed liner.

# 5.2.3 Fire Extinguishers and First Aid Kits

Fire extinguishers and first aid kits will be installed in the following locations:

- office building/ticket house/weigh station/maintenance shop
- selected on-site vehicles and equipment
- waste tires storage location

# 5.3 Accident or Injury

# 5.3.1 When an Injury Occurs

When an injury occurs, the following procedures will be implemented:

- shut down equipment in the immediate vicinity as is appropriate;
- determine extent of injuries (location, seriousness);
- if injuries are minor, administer First Aid, if trained, and notify Landfill Site Management;
- if injuries are serious, contact phone rescue squad (911) and provide the following:
  - · clearly state location, and
  - describe injuries;
- if victim is <u>not breathing and has a pulse</u>, administer <u>rescue breathing</u>, if trained in CPR while waiting for emergency response (911);
- if not trained, immediately notify facility personnel certified in administering CPR/First Aid; and
- notify Landfill Site Management.

#### **5.3.2** Procedures After an Accident

The following procedures will be implemented in the event of an accident:

- Accident Investigation The Landfill Site Management will make a complete investigation of the accident and events leading up to the time of the accident. The investigation will be started as soon as possible after the accident. All witnesses to the accident and persons involved in the accident will be interviewed.
- Determination of Cause After the facts about the accident have been gathered, the Landfill Site Management will make a determination as to the cause(s) of the accident.
- Filing of Reports The Landfill Site Management will complete and file the appropriate accident report forms.
- Corrective Steps After a thorough investigation and determination of the cause(s) of
  an accident, the Landfill Site Management will take corrective steps so that the same
  type of accident will not re-occur. These corrective steps may take the form of repair
  of faulty equipment, installation of safety equipment, or instruction of personnel in
  safe operating procedures.
- Discussion with Employees If it is determined that the cause(s) of the accident were related to employee work habits and that remedial safety instructions would be helpful, a meeting with site employees will be held. The accident and the corrective measures that will be taken will be discussed to prevent another accident. All employees will be instructed in proper safety procedures which should be followed.
- Follow-up The Landfill Site Management will follow-up the corrective measures to make certain that proper safety precautions are being taken. All unsafe practices will be called to the attention of the employees.

# 5.4 Release of Contamination to Environment (Remedial Response)

# 5.4.1 Response

If contamination is released to the environment, the following procedures will be implemented:

- determine location, extent, type, and, if possible, cause of release (e.g., leachate, contaminated surface water, fuel spill, etc.);
- notify Landfill Site Management and implement safety and emergency response procedures; and
- notify proper authorities including the Florida "Hot Line".

A list of individuals and emergency response agencies to contact in the event of a release of contamination to the environment is provided in **Attachment G**.

# 5.4.2 Follow-Up

Unless the occurrence of a contaminant release is clearly due to very unusual circumstances, the Landfill Site Management will take corrective action to prevent recurrence of the release. The corrective action will be approved by the FDEP.

A report will be filed at the landfill by the Landfill Site Management in order to have further reference for inquiries by authorities or WCOC personnel. The report will state:

- time/date of incident or its discovery
- type of release and effects
- source
- response and effectiveness
- agencies contacted
- corrective actions planned and scheduled

#### 5.5 Hazardous or Other Unauthorized Materials

In the event that a substance known to be or suspected of being hazardous or other unauthorized material is dumped from any vehicle at the waste disposal facility, the actions described below will be taken immediately.

#### 5.5.1 The Observer

#### 5.5.1.1 Known or Suspected Hazardous Material

The Observer will take the following actions:

- Immediately report the incident to the Landfill Site Management or their designee.
- Avoid exposure to the substance in question. Stay upwind.
- Observe where the material was dumped, by whom (which vehicle), how much was dumped, whether the container appears sound or is leaking, and what the substance looked and smelled like. Such observations will only be made with extreme caution and with the utmost regard for safety. DO NOT <u>SNIFF</u> OR TOUCH THE SUBSTANCE.
- Ask the individual who dumped the suspect load where the material was obtained.
- Isolate the approximate area of the suspected load before it is covered or mixed with wastes from other vehicles.

• Ask the driver of the vehicle to remain at the dumping point to ensure adequate vehicle identification. If the driver attempts to leave the discharge point, the observer should inform the Scale master and/or the Landfill Site Management.

#### 5.5.1.2 Other Unauthorized Material

When other unauthorized waste is discovered by a heavy equipment operator while working at the active face area, the heavy equipment operator will either immediately stop operation of the heavy equipment and remove the unauthorized waste to an isolated area away from the active disposal operations or will radio for other site personnel to assist in removing the unauthorized waste to the isolated location. Operations will resume upon removal of the unauthorized waste. All unauthorized waste staged in the isolated location will be removed at the end of the day and temporarily stored in a designated onsite location for proper management/disposal. Unauthorized putrescible waste will be removed from the site within 48 hours and non-putrescible waste within 30 days.

#### **5.5.2** Landfill Site Management

The Landfill Site Management will take the following actions if a known or hazardous material is discovered:

- Notify the FDEP.
- Record all pertinent facts regarding vehicle, including but not limited to: name of carting company; license plate number; where the load was obtained, if known; any visible evidence identifying the waste substance; and quantity and state of the substance (e.g., solid or liquid or if contained or loose).
- Maintain careful records of other costs incurred as a result of the dumping incident including, but not limited to, security costs in isolating the area, costs of removal (by contract or otherwise) of the suspect material, other costs of intermediate or ultimate treatment and/or disposal, and any other pertinent costs.
- Coordinate the removal of the unacceptable waste with the proper authorities.

# 5.5.3 Non-Discharged Load

If, before a waste load can be discharged (e.g., during inspection), it is discovered to contain, or is suspected of containing hazardous or other unauthorized materials, the same reporting procedures by the Observer and Landfill Site Management described for the discharged loads still apply, except concerning the discharging itself. In addition:

- inform the driver that his load is unacceptable and why;
- do not permit the load to be discharged; and

• suggest to the driver that he phone the FDEP to determine what he should do with the load.

# **5.6 Uncooperative Customers**

The following actions will be implemented if a customer will not obey site rules or cooperate with site personnel.

- if the customer is creating a substantial problem involving their or other's safety, or significantly interfering with disposal operations, the Landfill Site Management will decide what action should be taken;
- if the customer is creating a minor nuisance and does not respond to polite suggestions, the employee will record the vehicle description and license number, and report the incident to the Landfill Site Management or home office management; and
- in a case where a customer causes or threatens to cause harm to landfill property or personnel, or otherwise interferes with safe operation of the landfill, the Landfill Site Management will contact the Sheriff.

# 5.7 Inclement Weather

# 5.7.1 Operation in Wet Weather

<u>Problem</u>	Solution	
Saturated Unloading Area	1) Stockpile well-drained soil and apply as necessary.	
	<ol> <li>Keep compactors off area; use dozers on unloading area.</li> <li>Unload and push refuse perpendicular to area.</li> </ol>	
	3) Grade unloading area slightly to permit runoff.	
Mud Carried Onto Access/Public Roads	1) Carefully scrape mud from pavement.	
	2) Provide clean rock dressing to internal access roads. If internal access roads are properly maintained, then dirt on the tires of disposal vehicles will be thrown off prior to reaching public access roads.	
Cover is Wet/Unworkable	1) Maintain compacted, sloped stockpiles.	
	2) Use alternate cover approved by permit.	

# **5.7.2** Preparation for Inclement Weather

The following preparations will be made for inclement weather:

- Wet weather areas will be prepared during periods of dry soil conditions. The wet weather area will be constructed close to an all weather road. Work on the wet weather area will be performed at various times when personnel and equipment are not required for other higher priority assignments.
- Access roads around the site will be maintained as necessary. These roads will be
  maintained in a serviceable condition with the use of the available equipment on
  site, such as grader, water truck, dozer and loader. Major repairs will be scheduled,
  if required.
- Drainage structures, ditches, and sediment control will be checked to ensure they are in good condition and free of significant debris prior to anticipated heavy rains.
- Temporary (Operations Area) Drainage Control cover material, rock/sand, aggregate, and corrugated metal pipe, will be stockpiled for use in an emergency situation.
- When periods of high wind are predicted, litter fencing will be moved to close proximity of the working face and in the expected downwind direction. Cover may be required frequently during the day.

# **5.7.3** Preparation for Severe Weather or Hurricanes

In addition to the activities listed in Section 5.7.2 above, the following preparations will be made for severe weather or hurricanes. The following activities will be performed before and after the severe weather condition.

#### 72 hours prior to event

- Pick up any lose debris from the site
- Call supplier to top off all on-site fuel tanks
- Assess inventory
- Test generators
- Make final supply run for non-perishable food items and drinking water

#### 48 hours prior to event

- Continue with housekeeping efforts
- Perform administrative building flood prevention (to protect documents, equipment, furniture, etc.)
- Continue to evaluate situation pertaining to alternate sites

- Inform customers of expected service
- Supply written instructions to all employees (by WCOC human resource department)
- Allow employees time to secure their respective belongings

#### 24 hours prior to event

- Continue with housekeeping efforts
- Fit plywood shutters where necessary
- Distribute phone numbers to all employees
- Inform customers & corporations of possible cessation of operations

# 12 hours prior to event

- Secure the facility
- Fuel all vehicles and heavy equipment
- Park all track machines and rubber tire equipment close together (at ground level, i.e., off the top of the landfill)

#### Post - Event

- Key personnel (facility manager, operations manager, and certified operators) to assess site
- Contact all employees
- Initiate clean up/recovery of the facility
- Use of non-essential office staff for miscellaneous duties
- Resume limited or complete operations

## 5.8 Problems Affecting the Leachate Collection and Removal Systems

# 5.8.1 Interruption of Power Service to the Landfill

The ability to switch over to the secondary power supply allows the leachate collection and removal systems to continue operating with virtually no interruption. In the event that the main power service to the landfill is interrupted for more than 24 hours, the site will be switched over to the secondary power supply system consisting of diesel generators.

# 5.8.2 Interruption of Flow to Leachate Storage Facility

In the event that leachate flow to the leachate storage facility is temporarily interrupted, the leachate will be stored in the active cell(s). If the system cannot be restored within a reasonably acceptable period, leachate will be pumped directly from the sump to tanker trucks for off-site treatment.

## 5.8.3 Primary Leachate Sump Alarm Level Switch

An alarm level switch will be installed in one of the primary leachate sumps to notify the operator in the event that leachate levels in the sumps reach this level. The intent of the alarm is to notify the operator of a potential problem with the leachate pumps or piping. The alarm may indicate that either one or possibly both of the primary leachate pumps may have stopped working, the pumping capacity of both pumps has been exceeded, the storage containers are full, or there is possible blockage in the leachate transmission line. The operator shall observe the leachate pumps, pump control panels and flow meter to determine if either or both of the pumps are working. If at least one of the pumps is operational and there is no blockage in the leachate transmission line the operator will open the gate valve located in the secondary leachate manhole. By opening this valve leachate from the adjacent primary sump may flow into the secondary leachate sump for pumping. The operator shall record the flow meter reading on the secondary leachate sump pump prior to opening the gate valve. The operator shall also record the date and time of the occurrence and reason why the valve was opened (i.e., primary pumps failed, excessive leachate flow, etc.). The operator shall monitor the pumping of leachate to determine if the high leachate levels were associated with the pumps. The operator shall also examine the leachate transmission line manholes, piping and storage tanks assess any other potential problem. The leachate pumping system will require troubleshooting to determine the cause of the leachate build-up in the primary sumps and malfunctioning/inoperable pumps shall be replaced or repaired as soon as practical.

# **5.8.4** Managing Hazardous Leachate

In the event the leachate quality monitoring indicates the leachate is a hazardous material, the leachate will be managed in accordance with Chapter 62-730, F.A.C.

#### 6 SAFETY PLAN

#### **6.1** Emergency Procedures

- Posting of Procedures All emergency procedures (Emergency Contingency Plan Section 6 of this Operation Plan) will be updated as appropriate and after each emergency, if required. All emergency procedures will be posted in the Landfill Site Management's office, in conspicuous places at the site, and at the gate house.
- The name, location, and telephone number of the nearest doctors, medical treatment facilities, and ambulance services (contained in **Attachment G** of this plan) will be posted in the Landfill Site Management's office and all occupied buildings (i.e., maintenance building, gate house and office).
- Instructions on Procedures All new personnel will be instructed on the emergency procedures used at the landfill. All employees will be informed of any changes in emergency procedures.
- Responsibility of Employee It is the responsibility of every employee to know and remember their role in each emergency procedure at the site.

#### 6.2 General Safety Practices

- Knowledge of Procedures All employees at the landfill will know the proper procedures for reporting accidents, injuries, and fires.
- Posting of Information Roadway limits within the landfill footprint will be clearly
  posted as necessary. Site speed limits will be clearly posted on the main access road.
  Direction of travel and location of curves will also be posted. The location of
  disposal areas will be clearly indicated.
- Site User Rules Site user rules will be posted at the entrance to the landfill. Employees will watch for violations. Employees will explain rules to violators, stressing that the rules are for their protection. As a last resort, the Landfill Site Management will notify the County Sheriff's Office for further action.
- Discharging Loads For safe operations, the discharging area will be only slightly sloped (for drainage) at all times and equipment maintained in good repair.
- Safety Devices Proper safety devices, such as roll-over protective cabs, will be installed on all equipment and kept in good repair.
- Fire Extinguishers Fire extinguishers will be provided in buildings and on equipment. Each extinguisher will be appropriate for the types of fires likely and they should be checked or serviced as appropriate. Discharged (even partially) fire extinguishers will be removed and replaced with fully charged units.

- First Aid Kits First aid kits will be maintained in the main office building and in select site vehicles. An inventory of the first aid supplies should be maintained in order to re-supply the first aid kits when items used.
- Safety Meetings Safety meetings will be regularly scheduled. Situations that can
  cause accidents and ways to prevent them will be discussed. Also, the effectiveness
  of corrective actions following accidents at the site will be discussed.
- NO SMOKING will be allowed within the landfill area or near fuel storage facilities.

#### **6.3** Safety Equipment

Certain safety equipment is specified for equipment operator protection. It is the responsibility of every employee to ensure that their safety equipment is in good condition. All employees are to use their safety equipment at appropriate times. The safety equipment recommended for equipment operators is listed in Table 4.

## 6.4 <u>Site User Rules</u>

The following set of rules will be observed at the landfill.

- No Smoking Users will not smoke on the site.
- Children and Pets in Vehicles Individuals (children and pets) not involved in unloading refuse will remain in the vehicle.
- Persons Unloading to Remain Near Vehicle Persons unloading will remain within 10 ft (3 m) of their vehicle at all times.
- No one will be allowed to ride on the outside of a vehicle while on site.
- Discharge Waste Behind Vehicle Whenever possible, waste will be discharged immediately behind the unloading vehicle.
- Unloading No unloading by non-mechanized trucks or passenger cars is to be done using rapid acceleration or deceleration of the vehicle.
- Keep Tools in Vehicle Tools, removable tailgates, sideboards, wheelbarrows, ladders, and tarps will be kept in, on, or under the vehicles being unloaded to prevent damage to other vehicles or site equipment.
- Speed Limit The posted speed limit within the landfill site will be enforced.
   Operating personnel will direct users to further reduce their speed when justified by site conditions.
- No Scavenging Scavenging is not permitted at the landfill site.

- No Shooting Firearms are not permitted at the landfill site.
- No Explosives Explosives are not permitted at the landfill site.

#### 7 CLOSURE PLAN

# 7.1 <u>Introduction</u>

The purpose of this Closure Plan is to describe the general aspects of the landfill closure design for the JED facility. In general, the JED facility will be closed as sections of the landfill reach final design elevations (e.g. close-as-you-go) to minimize leachate generation in the landfill. The remainder of this Closure Plan describes (i) the final cover system design (Section 7.2); (ii) surface water drainage system (Section 7.3); (iii) final cover system components; and (iv) erosion minimization activities (Section 7.4).

#### 7.2 Final Cover System Design

The final cover system will include 3H:1V side slopes (between benches) with 15-ft wide benches every 40 vertical feet (at elevations 138, 178, 218, 258, and 298 feet, NGVD) and the top slopes will be graded at 5 percent.

# 7.3 Surface-Water Drainage System

Drainage swales are incorporated in the final cover system on the top and on the side slopes of the landfill. The drainage swales will convey water to the downdrains. The downdrains will convey the storm water runoff to the storm water retention basins at the toe of the landfill. The downdrains consist of corrugated HDPE pipes that tie into energy dissipater/junction boxes located at the toe of the waste slope.

As required, the swales, downchutes, culverts, and perimeter ditches will be maintained on a regular basis. Significant sediment and debris, which has accumulated in the swales, culverts, and perimeter ditches will be removed to facilitate flow and prevent overflow. Significant sediment and debris is considered any amount that impedes flow in the swale or any buildup greater than 0.5 feet.

## 7.4 Final Cover System Components

The cross section of the final cover system on the top slopes of the landfill is shown in the permit drawings and consists of, from top to bottom:

- a 0.5-ft (0.15-m) thick vegetative layer;
- a 1.5-ft (0.45-m) thick vegetative support layer;
- a 40-mil (1-mm) thick polyethylene (PE) geomembrane; and
- a 1-ft (0.3-m) thick intermediate cover layer.

The cross section of the final cover system on the side slopes of the landfill is shown in the permit drawings and consists of, from top to bottom:

- 0.5-ft (0.15-m) thick vegetative layer;
- a 1.5-ft (0.45-m) thick vegetative cover layer;
- a geocomposite drainage layer;
- a 40-mil (1-mm) thick PE geomembrane; and
- a 1-ft (0.3-m) thick intermediate cover layer.

The final cover system incorporates a geomembrane, which significantly reduces infiltration into the landfill cells. The grades of the final cover system are 3H:1V on the side slopes between benches and 5.0 percent on the top slopes.

#### 7.4.1 Vegetation

The surface of the final cover system will be vegetated either by seeding or sodding. The grass seed or sod will be Bahia, which has a high tolerance to drought. The contractor may use an alternate grass seed contingent upon proof that the grass is drought-resistant. The minimum requirements of the grass seed and sod are presented in the Technical Specifications on file with FDEP.

#### 7.4.2 Vegetative and Cap Protective Layer

The upper 6 inches of the final cover system will consist of loosely placed vegetative layer and will be vegetated to minimize erosion. The cap protective layer below the vegetative layer will consist of 18 inches of on-site soil (or approved equal). The cap protective layer will be compacted in the upper 6 inches during construction to inhibit root penetration into the drainage layer underlying the cap protective layer on the side slopes.

# 7.4.3 Geocomposite Drainage Layer

A geocomposite drainage layer consisting of a geotextile filter, a geonet drainage layer, and a geotextile friction layer will be placed beneath the cap protective layer on the 3H:1V side slopes. The geotextile filter, the geonet drainage layer, and the geotextile friction layer are bonded together to form the geocomposite drainage layer. The function of the proposed geotextile filter is to prevent soil particles of the overlying cap protective layer from penetrating and clogging the underlying geonet drainage layer. The purpose of the drainage layer is to remove the storm water reaching the geonet and to minimize the potential of pore water pressure build-up in the overlying cap protective layer. The purpose of the geotextile friction layer is to increase the interface friction between the geomembrane and the geonet and thereby increase the stability of the final cover system.

#### 7.4.4 Geomembrane

A geomembrane is proposed as a component of the final cover system to reduce infiltration of the storm water through the final cover system into the waste. The specified geomembrane is a 40-mil thick textured polyethylene geomembrane on the 3H:1V side slopes and a smooth polyethylene geomembrane on the 5 percent top deck surface. The texturing is necessary to increase the stability of the final cover system. Specified property values for the final cover geomembrane are provided in the Technical Specifications on file with FDEP. The specified geomembrane meets the requirements of Section 62- 701.600(3)(g), FAC.

#### 7.5 Erosion Minimization

Erosion of the final cover system will be minimized by final cover swales. The swales will intercept sheet flow from the final cover system. The final cover swales will direct the collected surface-water runoff to downchutes and the perimeter swale.

A vegetative cover will be placed on the final cover slopes of the landfill as described in Section 7.4.2. This vegetative cover will minimize erosion and reduce soil loss from the final cover system. The final cover system will be periodically inspected and erosion damage or vegetative stress observed during these inspections will be repaired before significant erosion has a chance to develop.

An initial watering schedule will be developed at the time of closure, and will be dependent on whether the landfill is closed in the dry season or the rainy season. The grass will be watered and fertilized, as necessary, to ensure continued growth.

#### 8 BORROW AREA OPERATIONS

# 8.1 Overview

Fill material needed for the JED facility construction and daily operations will be borrowed from excavations, or pits, located in the areas indicated on the Permit Drawings. Prior to any borrow activities in the location designated as Borrow Area A, the storm water management berm, in its interim configuration, will be constructed and vegetated. The outside toe of this berm will be constructed no closer than 25 feet to the nearest wetland boundary. In subsequent stages of the landfill development, the storm water management berm around Borrow Area A will be raised to its final height prior to the edge of the borrow excavation getting closer than 250 feet from the inside toe of the berm.

The development of Borrow Area C will be undertaken in future phases of the JED facility development. It is anticipated that this borrow area will be developed in a manner similar to Borrow Area A.

Two methods are proposed for excavating fill material from the borrow areas. These methods include: (i) mechanical excavation without dewatering (i.e. wet excavation) and/or (ii) dewatering the borrow area (i.e. dry excavation) and excavating fill using conventional earth moving equipment. Both methods are to be implemented in a manner which will minimize impacts to adjacent wetlands.

#### 8.2 Wet Excavation

Wet excavation is expected to be the primary method of borrow area operation during construction at the JED facility. This method of borrow area operation will require removal of soil materials from the pit without first dewatering the pit. Initially, the area will be cleared and grubbed and the topmost organic soil layers will be stripped and used for construction of the perimeter berm or stockpiled for future use. Next, typical excavation equipment such as a dragline or backhoe excavator will be positioned to remove soils and temporarily stockpile the material on the surface adjacent to the excavation. The slope of the temporary stockpile area will be maintained to channel excess water back to the open excavation or to allow infiltration. A bucket-loader or other suitable equipment will maintain the temporary stockpile and will load trucks or pans used to haul the material to the area of current construction or to designated stockpile areas.

All borrow areas will be developed from the center of the designated area towards the outer perimeter. The excavation equipment will continuously move around the perimeter of the borrow area excavation. After digging to the equipment's optimum depth, the equipment will move in a clockwise or counterclockwise direction to continually expand the pit until it reaches the final planned dimensions.

## 8.3 Dry Excavation

Dry excavation will be the alternative method of borrow area operation at the JED facility. This method of borrow area operation requires dewatering of the borrow area prior to removal of soil materials. Initially, the borrow area will be cleared and grubbed and the topmost organic soil layers will be stripped and used for construction of the perimeter berm or stockpiled for future use. Next, a ditch recharge system will be constructed between the area to be dewatered and adjacent wetland areas, which may be affected by the dewatering activities. The purpose of the ditch recharge system is to maintain a ground water level between the dewatered pit and the adjacent wetland, which will prevent detrimental affects to the wetland area. It is anticipated that WCOC may be required to obtain a water use permit from the South Florida Water Management District for the dewatering system if daily pumping quantities exceed 100,000 gallons. In conjunction with the water use permit application, a detailed layout of the recharge ditches, sequence of developing the dry pit, and location of pumps will be prepared.

The groundwater will be lowered in the borrow pit as the soil is excavated to provide trafficability in the excavation for equipment performing the excavation. It is anticipated that earth will be moved using self-loading pans, dump trucks loaded by backhoe, or other suitable heavy equipment, which will cycle through the borrow area to load and to the construction or stockpile site for unloading. As the excavation is progressively deepened, the ground water elevation in the excavation will be lowered ahead of the excavation bottom elevation. All water taken from the pit will be deposited in the recharge ditches, where it will maintain the ground water level in adjacent wetlands.

# TABLE 1 PERSONNEL REQUIREMENTS FOR RECEIPT OF UP TO 3,000 TONS OF WASTE PER DAY\* JED FACILITY

Personnel Classification	Total Number of Personnel Employed	Minimum Number of Personnel Required for Receipt of Waste
Office Administrator	1	0
Scale master	1	1
Landfill Equipment Operator (s)/Spotter(s)**	8	2
Landfill Site Manager/Operator***	1	1

## Notes:

<sup>\*</sup> For receipt of volumes greater than 3,000 tons of waste per day, site staffing will be increased accordingly.

<sup>\*\*</sup> Equipment Operators will perform spotting duties while operating heavy equipment

<sup>\*\*\*</sup> Random load waste inspector or designee

# TABLE 2 HEAVY EQUIPMENT REQUIREMENTS<sup>(1)</sup> FOR RECEIPT OF UP TO 3,000 TONS OF WASTE PER DAY<sup>(2)</sup> JED FACILITY

<b>Equipment On-site</b>	Quantities	
Chevy 1500 1/2 ton pick-up	1	
Caterpillar 12G Motor Grader	1	
CAT 613 Water Wagon	1	
6" water pumps	1	
4"water pumps	1	
Caterpillar D6 Dozer	1	
Volvo Articulating Hauling Truck	2	
Caterpillar 330 Excavator	1	
Caterpillar 836 compactor	2	

#### Note:

- (1) Equipment manufacturers' names are provided to indicate the approximate size and/or capacity of the equipment. The specific manufacturer for this equipment is not required but similar size is.
- (2) For receipt of volumes greater than 3,000 tons of waste per day, heavy equipment requirements will be increased accordingly.

# TABLE 3 HEAVY EQUIPMENT RENTAL COMPANIES JED FACILITY

Name of Rental Business	Phone Number
United Rental	(407) 332 - 1470
LundquistExcavating	(407) 847 - 9419

# TABLE 4 OPERATOR PROTECTIVE EQUIPMENT JED FACILITY

Equipment: Each piece of heavy equipment should be provided with:

Safety restraint belt

Roll-over bars

Backup warning system

Fire extinguisher

Personal: Equipment operators should have the following personal protective clothing

and accessories:

Ear muffs or ear plugs

Safety glasses or face shields

Rubber or leather (steel toe, shank) boots

Work gloves

Hard hats